THE IMPACT OF CULTURAL AND ORGANIZATIONAL FACTORS ON GENDER DIVERSITY IN LEADERSHIP: A GOVERNANCE PERSPECTIVE IN SAUDI PUBLIC ORGANIZATIONS

Vimala Venugopal Muthuswamy *, Jagathala Joghee Savithri **

* Corresponding author, Business Administration Department, King Faisal University, Hofuf, Saudi Arabia Contact details: Business Administration Department, King Faisal University, P. O. Box 400, Hofuf, Alhasa, 31982, Saudi Arabia ** Jansons School of Business, Coimbatore, India



How to cite this paper: Muthuswamy, V. V., & Savithri, J. J. (2025). The impact of cultural and organizational factors on gender diversity in leadership: A governance perspective in Saudi public organizations. Corporate Board: Role, Duties and Composition, 21(2), 45–57.

https://doi.org/10.22495/cbv21i2art4

Imps.//doi.org/10.22400/0542112d1

Copyright © 2025 The Authors

This work is licensed under a Creative Commons Attribution 4.0 International License (CC BY 4.0).

https://creativecommons.org/licenses/by/4.0/

ISSN Online: 2312-2722 ISSN Print: 1810-8601

Received: 10.09.2024

Revised: 18.12.2024; 19.03.2025; 02.05.2025

Accepted: 26.05.2025

JEL Classification: J16, M12, M14, J78, L32

DOI: 10.22495/cbv21i2art4

Abstract

Gender diversity in leadership is a crucial worldwide problem, especially inside Saudi public institutions, where cultural and organizational variables profoundly influence women's leadership prospects. This research analyzes the impact of cultural norms, leadership styles, and governance frameworks on gender diversity in leadership positions. Data were gathered from 138 participants in Saudi public organizations using a standardized survey. Descriptive statistics, correlation analysis, and regression models were used to evaluate the link among cultural views, leadership practices, and gender discrepancies. Research demonstrates that cultural attitudes and leadership styles significantly affect gender representation in leadership, with conventional male-dominated frameworks obstructing women's professional advancement. The findings underscore a significant association between gender diversity and leadership efficacy, highlighting the need for governance measures. This research the discourse on gender equality in leadership and offers ideas for promoting gender-inclusive leadership in Saudi public institutions. The subsequent study needs to investigate gender-specific difficulties within various sectors and assess the enduring effects of policy improvements.

Keywords: Gender Diversity, Leadership Styles, Cultural Norms, Organizational Barriers, Saudi Public Organizations

Authors' individual contribution: Conceptualization — V.V.M. and J.J.S.; Methodology — J.J.S.; Formal Analysis — J.J.S.; Investigation — V.V.M.; Resources — J.J.S.; Writing — V.V.M. and J.J.S.; Supervision — V.V.M.; Project Administration — V.V.M.; Funding Acquisition — V.V.M.

Declaration of conflicting interests: The Authors declare that there is no conflict of interest.

Acknowledgements: This work was supported through the Ambitious Funding track by the Deanship of Scientific Research, Vice Presidency for Graduate Studies and Scientific Research, King Faisal University, Saudi Arabia (Grant No. 242682).

1. INTRODUCTION

Gender diversity in leadership has emerged as a significant worldwide concern, with businesses endeavoring to cultivate inclusive workplaces that provide equitable opportunities for both men and women to succeed. Nonetheless, despite considerable advancements in fostering gender



equality via governmental programs and business regulations, women continue to be underrepresented in leadership roles throughout several locations, including Saudi Arabia (Syed et al., 2018; Alhassan & Al Doghan, 2022). The Kingdom's Vision 2030, a strategic framework for economic and social transformation, emphasizes female empowerment as a critical objective; nonetheless, cultural and organizational obstacles persist in hindering women's advancement to senior positions (Abadi et al., 2022). Despite legal advancements aimed at promoting gender inclusion, workplace dynamics often mirror deeply rooted cultural conventions and conventional gender roles, restricting women's access to leadership positions (Hodges, 2017; Alsharif, 2018).

A significant concern in this context is the disparity between corporate policies advocating gender equality and the enduring prejudices that obstruct women's professional progression. Studies demonstrate that leadership styles and behaviors significantly affect leadership effectiveness (McCleskey, 2014); yet there is a paucity of research on how these dynamics contribute to gender discrepancies in leadership roles within Saudi governmental organizations. Current indicates that male leaders are frequently linked to transactional leadership styles, which prioritize control and rewards, whereas female leaders are inclined to embrace transformational leadership methods that highlight collaboration, empowerment, and inclusive decision-making (Al-Shamrani, 2013). In Saudi Arabia, the efficacy of leadership, especially in public enterprises, is often influenced by hierarchical frameworks and culturally entrenched expectations, which predominantly endorse male-centric leadership models (Sian et al., 2020; Hodges, 2017). Cultural beliefs about women's leadership attributes, risk tolerance, and decisionmaking authority persistently influence the perception and treatment of female leaders in the workplace (Guy et al., 2008; Abalkhail, 2017).

This research seeks to examine the influence of cultural and societal factors on gender diversity in leadership roles, assess the contribution of leadership styles and behaviors to gender disparities in leadership, investigate the challenges women encounter in obtaining leadership positions within Saudi public organizations, and offer recommendations for fostering gender-inclusive leadership and enhancing organizational structures (Almalki et al., 2024). This research will provide significant insights into the current discussion on gender diversity in leadership by examining the following critical research questions:

RQ1: How do cultural and organizational factors affect gender diversity in leadership?

RQ2: Which leadership styles facilitate or obstruct gender inclusivity?

RQ3: What obstacles do women encounter in attaining leadership positions?

RQ4: How can governance frameworks be improved to promote inclusivity?

This study enhances the discourse on gender diversity in leadership, specifically examining the governance framework in Saudi Arabia, by investigating the impact of cultural and organizational variables on women's leadership chances. Comprehending these processes is crucial for formulating effective strategies that may tackle the gender discrepancy in leadership positions

(Husu et al., 2010). This study's findings will provide practical insights for policymakers, corporate executives, and human resource professionals aiming to improve gender diversity in leadership roles and align with Saudi Arabia's national transformation goals, thereby fostering inclusive governance and sustainable organizational growth (Al-Dajani & Alsahli, 2021; Riddle & Ayyagari, 2011).

This paper is structured as follows: Section 2 reviews the literature on gender diversity in leadership. Section 3 explains the research methodology. Section 4 presents the results. Section 5 discusses the findings, and Section 6 concludes with key insights and recommendations.

2. LITERATURE REVIEW

This section reviews the existing literature on gender diversity in leadership, focusing on the cultural, organizational, and societal factors that shape women's access to leadership roles, particularly within the context of Saudi public organizations.

The role of culture in shaping gender diversity in leadership has been extensively studied, with research highlighting how cultural values, norms, societal expectations impact women's opportunities and experiences in leadership positions. The experiences and access of women in leadership are significantly shaped by cultural ideas. van Emmerik et al. (2008) assert that individualistic cultures foster gender-inclusive leadership, while collectivist civilizations limit opportunities for women's leadership. Abadi et al. (2022) leadership. demonstrated that patriarchal institutions in the Middle East impede women's advancement in leadership due to prevailing social and institutional norms. Hodges (2017) investigated Saudi Arabia, where gender norms obstruct women's leadership. Guy et al. (2008) examined the influence of cultural norms on the emotional labor of women leaders in public service. Jamjoom and Mills (2022) analyzed Saudi women leaders reinterpret their how leadership experiences to surmount cultural constraints. Qureshi et al. (2011) investigated the influence of cultural norms in higher education and management on women's leadership, whereas Sian et al. (2020) analyzed the impediments to women's leadership in Saudi Arabia's audit organizations stemming from gender, politics, and religion.

Gender diversity, leadership style, and cultural variables influence leadership behaviors, and Alhassan and Al Doghan (2022) discovered that cultural expectations influence women's leadership opportunities. According to Chin (2013), leaders from various races, genders, and minority backgrounds are shaped by cultural norms pertaining to diversity. Syed et al. (2018) investigated gender equality in Saudi Arabia and found that cultural beliefs in the workplace hinder women's leadership potential. Abalkhail (2017) investigated the impact of societal and institutional constraints on Saudi women's aspirations and experiences in leadership within higher education. (2013)Varshney investigated Khan and leadership in Saudi transformational acknowledging that cultural norms may impede women while also enabling them to lead in distinct ways. Aladwey and Alsudays (2023) discovered that the cultural context affects corporate governance in Saudi Arabia, specifically regarding board gender diversity and corporate value. Abu Alsuood and Youde (2018) discovered that cultural norms significantly influence leadership practices among deans in Saudi universities. Hodges (2017) analyzed the entrenched cultural perspectives that hinder Saudi women from attaining leadership positions.

Research indicates that organizational barriers restrict Saudi women's opportunities for leadership in public institutions. Alsubaie and Jones (2017) investigated cultural and structural barriers to women's leadership in Saudi higher education. Abdelwahed et al. (2022) investigated Saudi women's entrepreneurship and identified that cultural norms and organizational frameworks impede women's leadership, particularly in both public and private sectors. Ely and Rhode (2010) emphasized societal and organizational prejudices against women's leadership capabilities. Hogan and Coote (2014) assessed Schein's notion of organizational culture and determined that male-dominated leadership paradigms impede women's advancement leadership roles, particularly in traditional Saudi Arabia. Hoyt and Simon (2024) observed that social psychological factors and gender biases exclude women from leadership positions, exacerbating organizational gender disparities. Lewis and Fagenson-Eland (1998) discovered that gender and organizational hierarchy influence perceptions of leadership conduct, with women seen as less competent at elevated levels. Rincón et al. (2017) assert that in Saudi state businesses, organizational frameworks that prioritize male leadership provide significant gender obstacles to attaining senior management positions.

Almalki et al. (2024) examined the healthcare education sector in Saudi Arabia and identified that cultural expectations and organizational structures women's leadership, particularly education. Parveen (2023) analyzed structural barriers to Saudi women's leadership, such as limited access to decision-making roles and the cultural perception of leadership as a maledominated domain. Al-Ahmadi (2011) analyzed the social, cultural, and institutional barriers to women's leadership in Saudi Arabia. Otayf (2024) investigated the subjective experiences of women leaders at Saudi government universities and found that gender biases and institutional support impede leadership. Al-Asfour et al. (2017) assessed the impediments to Saudi women's employment, including cultural norms and organizational practices that obstruct their professional advancement in both the public and private sectors. Alfarran et al. (2018) observed that gender-specific limitations and insufficient leadership development support impede the career opportunities of Saudi women. Alomair (2015) conducted a critical analysis of the literature about female leadership in higher education, revealing that Saudi women face institutional barriers, including inadequate mentorship and societal expectations, which restrict their leadership capabilities. Alzaaqi et al. (2025) investigated empowerment strategies within the Saudi health sector, identifying gender biases, inadequate leadership training, and insufficient governmental initiatives to support the advancement of women's leadership.

In addition to external barriers, gender diversity in leadership styles also influence how women are perceived and evaluated in leadership roles, with studies suggesting that women's

transformational leadership styles may be more effective in certain organizational contexts. Research indicates that male and female leaders differ in their leadership styles and effectiveness. Al-Shamrani (2013) discovered in Saudi Arabia that women in segregated workplaces exhibit more collaboration and transformative leadership, whilst males display more authoritarian or transactional styles. Angelakis et al. (2024) investigated crisis communication within German higher education and found that female leaders exhibited more empathy sentimentality in online interactions compared to their male counterparts, underscoring gender disparities in communication styles. Eklund et al. (2017) investigated leadership across several contexts and found that women generally exhibit more democratic and participative styles compared to men, who favor hierarchical and directive approaches. Górska (2016) posits that women are seen as more emotional and relationship-oriented, whilst men are regarded as more task-focused and directive, influencing leadership efficacy. Maher (1997) demonstrated that leadership stereotypes perpetuate gendered expectations of leadership success by linking women to transformational capabilities and men to transactional ones. (2014) observed that situational, McCleskev transformational, and transactional leadership styles may be beneficial in different contexts; nevertheless, gender may influence the selected style without affecting leadership efficacy.

Sabharwal et al. (2017) showed styles of Master of Public the leadership Administration (MPA) directors are influenced by gender, with women favoring collaborative and transformational approaches, while men go toward authoritarian forms. Jrasat and Zubaidi (2024) investigated Saudi women in healthcare, revealing that cultural norms and organizational frameworks impede their leadership performance; nevertheless, they also observed that female leaders exhibit more empathy and support than their male counterparts. Sobaih and Abu Elnasr (2024) analyzed the Saudi tourist sector and the obstacles women have in overcoming the glass barrier. Research indicates that women's leadership styles in this industry are more inclusive and relationship-focused; nonetheless, they gender-based encounter considerable obstacles. Dahlan (2024) examined leadership development in Saudi higher education institutions, noting that although women leaders often use collaborative and transformational strategies, gender biases impede succession planning and professional advancement. Ur Rehman and Alorifi (2024) discovered that Saudi women entrepreneurs choose transformational leadership, which prioritizes innovation and change; nevertheless, societal expectations and gender biases impede their leadership capabilities. Aldekhyyel et al. (2024) investigated Saudi women's perspectives healthcare leadership and discovered a preference for participatory leadership styles, which align with the objectives of Saudi Vision 2030's health reform; nonetheless, they continue to face challenges in the male-dominated healthcare industry. Addin participation in (2025)examined women's the arbitration of administrative contract disputes and how gender disparities hinder women's leadership across several domains, obstructing their professional advancement.

The impact of governance frameworks and policies on gender diversity in leadership is particularly relevant in the Saudi context, where national initiatives and institutional policies are designed to promote gender equality but still face significant cultural resistance. Abdallah (2024) investigated the influence of gender, cultural, and psychological factors on the formation of leadership in distance education, concluding that these components shape women's leadership styles and opportunities within the educational sphere. Alshammari et al. (2020) discovered that Saudi nursing leaders use emotional intelligence and genuine leadership; yet, organizational regulations and gender roles impose constraints on them. Groves and LaRocca (2011) investigated ethical leadership in business, revealing that ethical leadership styles, principles and such leadership, transformational and transactional but influence corporate governance, prejudices impede women's ascension to leadership roles, van Emmerik et al. (2008) examined global leadership practices and discovered that cultural and gender factors influence leadership efficacy, with Saudi Arabia's cultural norms promoting gender diversity in leadership positions. Husu et al. (2010) examined the reinforcement of gendered leadership stereotypes by organizational policies and the divergent perceptions of leadership held by men and women. Pick (2024) analyzed the impact of governance structures and regulations on women's access to power and leadership roles across many sectors, including the Middle East. Riddle and Ayyagari (2011) analyzed the corporate ethics of female Egyptian managers, which may be relevant to Arabia, where cultural norms organizational limitations provide similar gendered leadership challenges.

Despite advancements in organizational policies and diversity management in Saudi Arabia, Abaker et al. (2019) identified that gender-specific regulations and practices continue to hinder women's attainment of leadership roles. Alhejji et al. (2018) investigated gender equality methods at a British multinational corporation in Saudi Arabia and found that formal policies and informal cultural norms influence gender equality, with informal distance often perpetuating gender disparities. Aravssi (2020)examined Saudi socioeconomic advancement, its Gender Inequality Index, and recommendations for improving gender equality in leadership, notwithstanding traditional roles. Alshareef and Sandhu (2015) integrated corporate social responsibility into corporate governance, demonstrating that board diversity, particularly in gender, enhances governance and social responsibility within Saudi firms. In Gulf Cooperation Council countries such as Saudi Arabia, board gender diversity is positively correlated with corporate social engagement; yet, cultural norms hinder gender equality in leadership roles. Issa and Fang (2019) discovered that gender diversity on corporate boards enhances corporate social responsibility in Arab Gulf countries. Chebbi and Ammer (2022) discovered that in Saudi Arabia, board composition and governance structures may enhance environmental, social, and governance disclosures, with gender diversity serving as a moderating factor. Doumato (1992) analyzed the impact of the Saudi Arabian monarchy's policies on women's roles in public and leadership positions.

Eichenauer and Ryan (2024) discovered that incongruent gender leadership expectations resulted in women being evaluated more critically than men. Women's activities are often seen as less effective than men's, which impedes their leadership development. Itzkovitch-Malka (2024) discovered that gendered expectations for leadership positions intensify during crises, with women often facing scrutiny and skepticism regarding their leadership capabilities. A meta-analysis conducted by Badura et al. (2018) showed that gender-based biases associating leadership qualities with masculinity diminish the likelihood of women assuming leadership roles in competitive environments. Tremmel and Wahl (2023) discovered that male leaders exhibit decisiveness and assertiveness, whereas female leaders have caring qualities but possess inferior decision-making skills, hence limiting their leadership opportunities. The 2024 song juxtaposes civilizations' ideals toward gender equality with the advancement of women's leadership, illustrating that certain cultures impose more pronounced gender standards than others.

Algahtani (2021) investigated the challenges faced by women in leadership within Saudi higher education, highlighting cultural norms and institutional barriers that limit their opportunities. Alsharif (2018) identified organizational barriers and gendered expectations that hinder women's ascension to leadership positions at Saudi public universities. Alwazzan and Rees (2016) investigated the medical education of Saudi women and found that gendered biases, including perceptions of diminished capability in leadership roles, impede their professional progression. Bakr investigated the empowerment of Saudi women in academic leadership roles and discovered that cultural norms and biases around gender roles in education continue to obstruct women's leadership. Thelma and Ngulube (2024) observed institutional and cultural biases, particularly the belief that women are unsuitable for leadership, impede the global advancement of women's leadership. Tabassum and Nayak demonstrated that leadership style preconceptions, which link leadership qualities to masculinity, hinder women's opportunities for leadership from a management perspective. Setyaningrum and Juansih (2024) compiled data on gender-related challenges in women's leadership careers, including gender biases, cultural expectations, and insufficient mentorship as obstacles to women's leadership.

To tackle these difficulties, research promotes gender-inclusive leadership models organizational changes that provide equitable opportunity for women, especially in public sector leadership positions, which have historically been male-dominated. Ehrhart and Klein (2001)investigated charismatic leadership the influence of followers' ideas and personalities on their decisions, emphasizing the need for inclusive leadership in fostering diverse leadership styles. Burke and Collins (2001) examined gender diversity in leadership styles and managerial competencies, advocating for organizations to prioritize both masculine and feminine leadership qualities to enhance diversity and inclusion. Itzkovitch-Malka (2024) investigated the evolution of gender biases related to leadership positions during crises, highlighting the need for strategies to eradicate these prejudices and facilitate effective leadership for both men and women. Gooty et al. (2023) called for gender-inclusive leadership in business schools to provide fair opportunities for women to attain leadership positions. Najmaei (2018) analyzed the strategic leadership paradigm and promoted gender-inclusive leadership methodologies to foster an equitable workplace culture. Billah and Kabir emphasized the need for inclusive governance frameworks that enable women in leadership roles to improve the public sector in Bangladesh. Purwaningtias and Purba inclusive discovered that gender influences leadership behaviors that challenge corporate norms for the benefit of society. Kamalizeni et al. (2018) analyzed the underrepresentation of women in public enterprises in Swaziland and suggested gender-inclusive organizational strategies enhance women's leadership. Akbar et al. (2023) assessed women's leadership within Saudi Arabia's emphasized education sector and for leadership the requirement development programs in mostly male academic institutions.

Chung and Grichting Solder (2024) proposed the use of European concepts to develop genderinclusive urban environments in the Arabian Gulf, aimed at advancing female equality in public space governance. Aldossari and Calvard (2022) analyzed the dynamics of resistance and feminism inside Saudi Arabian institutions and championed female in leadership roles. According Chalermchaikit et al. (2024), inclusion in human resource management may improve female participation in leadership within industries such as hospitality. Alshdiefat et al. (2024) advocated for legislative and organizational reforms to remove barriers to women's leadership in Jordanian colleges. Proff and Musalam (2024)emphasized the importance of supportive networks mentorship in addressing the obstacles of educational leadership for women. Al-Dajani and Alsahli (2021) assessed the challenges faced by Saudi women entrepreneurs and recommended institutional enhancements to facilitate women's leadership in the entrepreneurial domain. Ottmann (2024) emphasized the need for gender-inclusive leadership in the architecture, engineering and construction sectors to achieve long-term social and environmental goals and sustainable development. Xu et al. (2024) discovered that gender-inclusive leadership techniques might promote public sector expansion and societal progress. Al-Lamky (2007)articulated the perspectives of Omani female leaders about the feminization of Arab leadership to cultivate more inclusive and effective systems.

The current research emphasizes the influence of cultural and organizational determinants on gender diversity in leadership; nevertheless, there is a notable deficiency in comprehending how these elements precisely interact within Saudi public institutions. Prior research has examined the impact of cultural norms and societal expectations, including patriarchal frameworks (Abadi et al., 2022) and conventional gender roles (Hodges, 2017), on the restriction of women's leadership opportunities; however, there is a lack of emphasis on how these factors influence leadership styles and efficacy within public organizations. Despite the extensive documentation of gender disparities in leadership styles, wherein women frequently employ more transformational collaborative and methods (Al-Shamrani, 2013), the precise influence of these disparities on leadership efficacy within Saudi public sectors, particularly in male-dominated domains such as healthcare and academia, remains inadequately examined (Alwazzan & Rees, 2016; Jrasat & Zubaidi, 2024). Moreover, while research has identified organizational obstacles, gendered policies and prejudices (Alsubaie & Jones, 2017; Al-Ahmadi, 2011), there is little investigation into how these barriers explicitly combine with influence cultural expectations to advancement in leadership roles. The need for gender-inclusive leadership (Gooty et al., 2023) and governance changes (Kamalizeni et al., 2018) is widespread, although practical methods specific to the Saudi context are few. This study seeks to address these deficiencies by analyzing the cultural and organizational determinants that affect gender diversity in leadership across Saudi institutions and offering specific suggestions to promote a more inclusive leadership framework.

3. RESEARCH METHODOLOGY

This study utilizes qualitative research methods to thoroughly investigate the impact of cultural and organizational variables on gender diversity in leadership positions within Saudi public organizations. technique The incorporates descriptive statistics, correlation analysis, and regression analysis to provide a comprehensive assessment of the correlations among important factors, such as cultural influences, gender disparities, and leadership behaviors (McCleskey, 2014).

3.1. Research design

A descriptive research approach is used, enabling a comprehensive analysis of gender diversity in leadership across Saudi public institutions. This methodology is appropriate for gathering both qualitative and quantitative data, allowing a thorough study of the elements leading to gender discrepancies in leadership positions (Groves & LaRocca, 2011). Descriptive statistics capture essential features of the sample, while correlation and regression analysis evaluate linkages and predicted variables affecting gender diversity in leadership (Ely & Rhode, 2010; Alsubaie & Jones, 2017).

3.2. Participants and sampling

The research included 150 participants from several Saudi public institutions, all occupying leadership roles at various hierarchical tiers. The sample includes people from first-level, middle, senior, and top management, guaranteeing comprehensive coverage of leadership experiences. To mitigate bias, random stratified sampling was used, guaranteeing a proportional representation of respondents from various management tiers (Lewis & Fagenson-Eland, 1998; Rincón et al., 2017). This method guarantees the research encompasses a wide array of viewpoints from sectors like healthcare, education, and public administration (Almalki et al., 2024).

A total of 138 valid answers were obtained, resulting in a 92% response rate, deemed sufficient for statistical analysis (Abaker et al., 2019). The gender distribution of participants indicates 66.7% male and 28.3% female, highlighting significant gender discrepancies in leadership

positions in Saudi Arabia (Hodges, 2017; Sian et al., 2020). Furthermore, 37.7% of participants were from mid-level management, while 32.6% were from middle management, suggesting the presence of career advancement obstacles at these tiers, especially for women (Alqahtani, 2021; Alsharif, 2018).

3.3. Data collection

Data were collected by an online questionnaire disseminated via professional networks, institutional emails, and social media platforms (Jizi et al., 2022). The questionnaire included explicit instructions, an opening statement detailing the study's aims, and Likert scale questions assessing cultural views, leadership styles, gender attitudes, and leadership experiences (Al-Shamrani, 2013). The Likert scale spanned from "strongly agree" (5) to "strongly disagree" (0), providing a consistent method for assessing respondents' opinions toward leadership and gender diversity (Alhejji et al., 2018).

The questionnaire included demographic inquiries on gender, age, job title, years of experience, and employment sector, allowing a comprehensive knowledge of participant profiles and their significance to leadership diversity (Abalkhail, 2017; Abu Alsuood & Youde, 2018).

3.4. Data analysis

Quantitative data were examined by descriptive statistics to evaluate the mean and standard deviation (SD) of major variables, including respondents' perspectives on leadership and gender diversity (Song, 2024). Correlation analysis was used to discern links among cultural influences, leadership practices, and gender inequalities, facilitating the evaluation of the relevance of cultural and organizational elements in determining leadership chances for women (Aladwey & Alsudays, 2023; Syed et al., 2018).

A regression analysis was conducted to examine how leadership dimensions and cultural perceptions predict gender disparities in leadership positions. The regression model is presented by the independent and dependent variables.

Independent variables (predictors):

- culture shapes beliefs and perceptions about leadership and gender roles;
- interpersonal leadership skills and their impact on leadership effectiveness;
- leadership style (transformational vs. transactional);
- leadership behavior, including decision-making, empowerment, and communication.

Dependent variable:

• gender diversity in leadership positions (assessed through representation, career progression, and leadership effectiveness).

The regression model was tested for significance (analysis of variance (ANOVA) test: F-statistic = 8.698, p < 0.001), confirming that cultural and leadership variables significantly predict gender disparities in leadership roles (Groves & LaRocca, 2011; Riddle & Ayyagari, 2011).

3.5. Reliability and validity

Cronbach's alpha test was used to assess the reliability and validity of the survey instrument, evaluating internal consistency¹. The test yielded an alpha value of 0.8, indicating robust dependability since values beyond 0.7 are deemed appropriate for research projects (Abaker et al., 2019). A pilot study was done to develop survey questions, ensuring clarity in phrasing and structure in alignment with research goals (Alomair, 2015; Alzaaqi et al., 2025).

3.6. Alternative research methods

While the mixed-methods approach provides a balanced analysis, alternative methodologies could have been considered. Ethnographic research could have provided deeper qualitative insights into how cultural norms impact leadership interactions (Husu et al., 2010). Longitudinal studies tracking gender diversity over time could offer insights into evolving leadership dynamics and policy effectiveness (Pick, 2024). Furthermore, in-depth qualitative interviews with Saudi women leaders could supplement the quantitative findings with richer, personal experiences (Bakr, 2022).

3.7. Ethical considerations

This study complies with rigorous ethical standards, guaranteeing informed consent, data confidentiality, and voluntary participation (Chalermchaikit et al., 2024). Participants were apprised of the study's objective, and data confidentiality was maintained throughout the research procedure. Furthermore, bias mitigation strategies were used in the survey design and data collection to reduce researcher-induced effects (Alhejji et al., 2018).

4. RESULTS

Table 1 presents the demographic profile of the respondents, which provides significant insights into the composition of the sample. In terms of age, the majority of respondents fall within the 26-30 years old range, accounting for 31.9% of the total, followed by 31-35 years old with 27.5%, and 36+ years old at 24.7%. The smallest group is the 21-25 years old age group, representing 15.9% of the respondents. This indicates that the sample is largely composed of younger to mid-career professionals. Regarding seniority, 37.7% of respondents occupy mid-level positions, while 32.6% hold middle-management roles. The first-level management category comprises 15.9% and senior management makes up 13.8%, suggesting a higher concentration of middle-tier leadership within the sample. In terms of gender, the study shows a notable imbalance, with 66.7% male and 28.3% female respondents, highlighting a predominance of male leaders in the sample. When considering work experience, 29% of respondents have less than 5 years of experience, while 26.1% have 5-10 years. 24.3% have 11-15 years, and 20.6% have 16+ years. This distribution indicates that the sample is relatively young, with many respondents still early in their careers. Educationally, the majority of

¹ https://stats.oarc.ucla.edu/spss/faq/what-does-cronbachs-alpha-mean/



respondents hold a Bachelor's degree (54.3%), while 29% have a Master's degree and 16.7% hold a Doctoral degree, reflecting a well-educated sample. As for the organizational sector, 55.8% of the respondents work in the private sector, while 28.3% are in the public sector, suggesting that private organizations are more heavily represented. Finally, the respondents are predominantly from the Eastern region, with 79.7% of the sample residing there, indicating a regional concentration that may influence leadership perceptions and practices. This demographic distribution helps to contextualize the findings of the research, particularly regarding diversity, leadership styles, gender and organizational dynamics in Saudi public organizations.

Table 2 presents the descriptive statistics (mean and SD) for various gender diversity and leadership variables, reflecting respondents' views on how cultural influences, gender diversity, and leadership styles impact women's leadership roles. The mean scores provide insight into the central tendency of responses, while the SD reflect the variation in those responses.

Table 1. Demographic variables of respondents

Variable	Category	Percentage (%)	
	21-25 years old	15.9	
Ago group	26-30 years old	31.9	
Age group	31-35 years old	27.5	
	36+ years old	24.7	
	First-level	15.9	
Seniority position	Mid-level	37.7	
Semority position	Middle	32.6	
	Senior	13.8	
Organizational	Private	55.8	
sector	Public	28.3	
Gender	Male	66.7	
Genuei	Female	28.3	
	< 5 years	29	
Work experience	5-10 years	26.1	
WOLK EXPELLENCE	11-15 years	24.3	
	16+ years	20.6	
	Bachelor's degree	54.3	
Education level	Master's degree	29	
	Doctoral degree	16.7	
Living region	Eastern	79.7	

Source: Authors' elaboration.

Table 2. Descriptive statistics for gender diversity and leadership variables: Mean and standard deviation

Variable Variable	Mean	SD
Cultural influences impacting gender diversity	3.5	1.05
Opportunities for women leaders in the workplace are proportionally affected by cultural influences	3.6	1.05
Women leaders have developed overall leadership competencies in different sectors	3.4	1.0
Culture shapes beliefs and perceptions about leadership	3.3	0.9
Culture and experience of women's leadership roles affect their access to leadership positions and career advancement	3.5	1.05
Leadership styles standardized by industry leaders have great importance on the role of women in leadership positions	3.0	1.1
Gender diversity should be considered while offering higher-level managerial positions	3.1	1.1
Gender diversity impacts interpersonal leadership skills and style	3.3	1.0
Gender diversity limits the assignment of certain tasks, impacting leadership development skills	3.6	1.05
Ethical challenges in providing opportunities to women in leadership positions	3.3	1.1
Women in unprecedented leadership positions significantly impact global women's empowerment rate	3.4	1.05
There is a significant gender equality gap in leadership positions	3.5	1.05
Motivational leadership style and personality affect gender diversity and leadership behavior	3.0	1.1
Gender diversity has no effect on perceived leadership behavior and overall management effectiveness	2.6	1.2
Leadership style and job satisfaction are highly impacted by gender diversity	3.1	1.05
Gender inequality exists in various sectors	3.4	1.1
Women are second to men in various business sectors	3.5	1.1
Women's freedom to apply administrative style is restricted by traditional styles standardized by society	3.6	1.1
Behavioural gender diversity is directly linked to women being in lower leadership positions	3.5	1.1
Leadership style is correlated with management skills	2.9	1.15

Source: Authors' elaboration.

The first variable, cultural influences impacting gender diversity, shows a mean of 3.5 and a SD of 1.05, indicating moderate agreement with the idea that cultural influences affect gender diversity. However, the relatively high SD suggests that there is some variation in respondents' views. Similarly, the statement regarding opportunities for women leaders in the workplace being proportionally affected by cultural influences also received a high mean of 3.6 and an SD of 1.05, indicating strong agreement but with moderate variation. This suggests that while many respondents agree that cultural factors influence women's opportunities, there are differing levels of agreement.

Regarding women leaders' competencies across sectors, the mean of 3.4 with an SD of 1.0 reflects moderate agreement and indicates relatively consistent responses, signaling that respondents largely agree that women leaders have developed the necessary competencies, though opinions vary slightly. The variable of culture shaping beliefs and perceptions about leadership yielded a mean of 3.3 and a SD of 0.9, which shows moderate agreement

with the statement and relatively low variation, meaning there is a consensus among respondents regarding the role of culture in shaping leadership perceptions.

On the issue of culture and women's leadership experiences affecting career advancement, the mean is 3.5 with an SD of 1.05, suggesting a moderate agreement and diverse opinions on how these factors impact women's access to leadership roles. The item regarding leadership styles standardized by industry leaders received a mean of 3.0 and an SD of 1.1, which indicates more mixed responses, with some respondents neutral or disagreeing, and others agreeing with the importance of standardized leadership styles in affecting women's roles.

Respondents also expressed moderate to strong agreement with the statement that gender diversity should be considered in offering higher-level managerial positions (mean 3.1, SD 1.1). However, as reflected in the higher SD, there was significant variability in opinions about whether gender should be a factor in such decisions. Similar results were seen in the statement about gender

diversity impacting interpersonal leadership skills and styles (mean 3.3, SD 1.0) and limiting leadership development through task assignments (mean 3.6, SD 1.05). Both of these indicate agreement with the idea that gender diversity influences leadership behaviors, with some variability in the responses.

The statement about ethical challenges in offering women leadership opportunities garnered a mean of 3.3 and an SD of 1.1, showing moderate agreement with varying opinions on the ethical surrounding gender and leadership. Regarding the influence of women in unprecedented global leadership positions on women's empowerment, respondents agreed with a mean score of 3.4 and an SD of 1.05, again showing moderate agreement but with slight variations in responses. There is also moderate agreement on the presence of a gender equality gap in leadership positions (mean 3.5, SD 1.05).

Regarding motivational leadership styles and personality impacting gender diversity and leadership behavior, the mean of 3.0 and SD of 1.1 suggest a neutral to moderate agreement, with considerable variation in responses. However,

respondents largely disagreed with the idea that gender diversity have no effect on perceived leadership behavior and overall management effectiveness, as reflected in a lower mean of 2.6 and a higher SD of 1.2.

Subsequent inquiries about gender inequality in various sectors (mean 3.4, SD 1.1) and women being second to men in business sectors (mean 3.5, SD 1.1) show strong agreement with these concerns, though there is still some variation in responses. The statement about women's restricted freedom in applying administrative styles due to societal norms received the highest mean score of 3.6, indicating strong agreement, though responses varied to some degree, as shown by the SD of 1.1. Similarly. behavioral gender diversity being linked to lower leadership positions for women showed strong agreement (mean 3.5, SD 1.1). Finally, the statement about the correlation between leadership style and management skills had the lowest mean of 2.9 and the highest SD of 1.15, indicating a disagreement with the notion that leadership style is always correlated with management skills, with significant variation in respondents' opinions.

Table 3. Correlation matrix of cultural influences, gender diversity, and leadership variables

Variables	CI	GD	LS	TA	LSK	WUL	LE	EMP
CI	1							
GD	0.62 (0.000)	1						
LS	0.56 (0.001)	0.55 (0.002)	1					
TA	0.54 (0.003)	0.58 (0.001)	0.59 (0.000)	1				
LSK	0.53 (0.004)	0.57 (0.002)	0.60 (0.000)	0.63 (0.000)	1			
WUL	0.59 (0.000)	0.63 (0.000)	0.68 (0.000)	0.65 (0.000)	0.66 (0.000)	1		
LE	0.50 (0.003)	0.62 (0.000)	0.67 (0.000)	0.61 (0.001)	0.69 (0.000)	0.75 (0.000)	1	
EMP	0.61 (0.000)	0.65 (0.000)	0.70 (0.000)	0.64 (0.000)	0.67 (0.000)	0.78 (0.000)	0.80 (0.000)	1

Note: Cultural influences (CI), Gender diversity (GD), Leadership styles (LS), Task assignments (TA), Leadership skills (LSK), Women in leadership (WUL), Leadership effectiveness (LE), Empowerment (EMP).

Source: Authors' elaboration.

The correlation matrix in Table 3 reveals significant relationships between cultural influences, gender diversity, and various leadership variables. Cultural influences show a strong positive correlation with gender diversity, leadership styles, task assignments, and women in leadership roles, highlighting the powerful role that culture plays in gender-based leadership shaping dynamics. Specifically, cultural influences have a significant correlation with gender diversity (0.62, Sig. 0.000), suggesting that cultural norms significantly impact gender roles in leadership positions. Additionally, cultural influences are strongly linked to leadership Sig. 0.001), styles (0.56,task assignments Sig. 0.003), and women in leadership (0.54,(0.59, Sig. 0.000), emphasizing how culture impacts leadership approaches and the participation of women in leadership positions. Gender diversity themselves exhibits a strong correlation with leadership styles (0.55, Sig. 0.002), task assignments (0.58,Sig. 0.001), and leadership (0.57, Sig. 0.002), indicating that gender influences leadership is the way enacted the development leadership competencies. of Moreover, gender diversity also correlates positively with leadership effectiveness (0.62, Sig. 0.000) and empowerment (0.65, Sig. 0.000), suggesting that gender disparities in leadership roles directly affect both the perceived effectiveness of leaders and the empowerment of individuals in the organization. Women in leadership positions show a particularly strong correlation with leadership effectiveness (0.75, Sig. 0.000) and empowerment

(0.78, Sig. 0.000), reinforcing the idea that increasing female representation in leadership roles significantly boosts both empowerment and leadership outcomes. Empowerment itself also demonstrates strong correlations with leadership effectiveness (0.80, Sig. 0.000) and women in leadership (0.78, Sig. 0.000), confirming the integral role empowerment plays in enhancing leadership quality. The correlations in Table 3 indicate that cultural influences and gender diversity are intricately linked to multiple aspects of leadership, especially regarding empowerment and leadership effectiveness, with culture significantly shaping leadership behaviors and opportunities for women.

Table 4. Model summary: Composite measure of leadership dimensions

Model	R	R-square	Adjusted R-square	Std. error of the estimate
1	0.705	0.495	0.460	0.870

Note: Predictors: (Constant), a composite measure of leadership dimensions (Culture shapes beliefs and perceptions, Interpersonal leadership skills, Leadership style, Leadership behavior). Dependent variable: Gender diversity in leadership position.

Source: Authors' elaboration.

Table 4, the model summary table, presents an overview of the regression model's overall fit, which seeks to elucidate the variance in gender disparities in leadership positions through a composite measure of leadership dimensions as predictors, including cultural influences on beliefs

and perceptions, interpersonal leadership skills, leadership style, and leadership behavior. An R-value of 0.705 indicates a moderate to high positive correlation between the independent variables and dependent variable, indicating the predictors together account for a substantial percentage of the variation in gender disparities in leadership roles. The model, with an R² of 0.495, accounts for 49.5% of the variability in the dependent variable, indicating a moderate degree of explanatory power. The adjusted R2 of 0.460, which accounts for the number of predictors, indicates that about 46% of the variance is explained after considering the model's complexity, hence affirming the significance of the predictors.

Table 5. ANOVA: Composite measure of leadership dimensions

Model 1	Sum of squares	df	Mean square	F	Sig.
Regression	53.734	6	8.956	8.698	0.000
Residual	117.373	132	1.030		
Total	171.107	138			

Note: Predictors: (Constant), a composite measure of leadership dimensions (Culture shapes beliefs and perceptions, Interpersonal leadership skills, Leadership style, Leadership behavior). Dependent variable: Gender diversity in leadership position.

Source: Authors' elaboration.

Table 5 presents the analysis of variance for the regression model. The F-statistic is 8.698, accompanied by a p-value of 0.000, indicating the statistical significance of the model. This indicates that the variables used in the model (Cultural influences on attitudes and perceptions, Interpersonal leadership competencies, Leadership style, and Leadership conduct) combined account for a substantial amount of the variation in gender disparities in leadership roles. The sum of squares for the regression is 53.734, indicating the variance elucidated by the model, but the residual value of 117.373 signifies the unexplained variance. The entire sum of squares, indicating the overall variation in the dependent variable, is 171.107. The mean square values for regression and residuals are 8.956 and 1.030, respectively. The significance value of 0.000 validates the statistical significance of the model.

Table 6. Coefficients: Composite measure of leadership dimensions

Model 1	Unstandardized coefficients		Standardized coefficients		
	Beta	Std. error	Beta	t	Sig.
(Constant)	0.305	0.381		0.801	0.425
Culture shapes beliefs and perceptions	0.327	0.105	0.306	3.136	0.013
Interpersonal leadership skills	0.102	0.120	0.066	0.680	0.450
Leadership style	0.297	0.125	0.248	2.600	0.009
Leadership hehaviour	0.112	0.099	0.098	1.089	0.268

Note: Dependent variable: Gender diversity in leadership position.

. Source: Authors' elaboration.

Table 6, the coefficients table, delineates the correlation between each predictor and gender disparities in leadership roles. The constant has a B-value of 0.305; nevertheless, the p-value of 0.425 indicates a lack of statistical significance. Culture influences attitudes and perceptions, with a B-value of 0.327 and a p-value of 0.013, indicating a substantial beneficial impact on gender disparities in leadership roles. Interpersonal leadership abilities have a B-value of 0.102; nevertheless, the p-value of 0.450 indicates that they do not substantially influence gender disparities in leadership roles. The leadership style exhibits a B-value of 0.297 and a p-value of 0.009, indicating a substantial beneficial impact on gender disparities in leadership roles. The leadership behavior has a B-value of 0.112; nevertheless, the p-value of 0.268 indicates a lack of meaningful influence.

5. DISCUSSION

This research offers essential insights the impact of cultural and organizational variables on gender diversity in senior positions within Saudi governmental organizations. The demographic study indicates a notable gender imbalance, with male respondents constituting 66.7% and female respondents accounting for just 28.3%. This disparity reflects wider socioeconomic patterns, whereby males persist in occupying leadership roles despite the increasing availability of a highly educated and competent female workforce (Syed et al., 2018; Alhassan & Al Doghan, 2022). The disproportionate representation of mid-level (37.7%) and middle management (32.6%) respondents indicates that institutional barriers, especially those associated with organizational hierarchies and leadership expectations, may impede the career progression of women in public organizations (Alsubaie & Jones, 2017; Sian et al., 2020).

Descriptive statistical research the significant influence of cultural variables on diversity in leadership. The gender respondents demonstrated a moderate to strong agreement with the impact of cultural norms on women's leadership chances, yielding a mean score of 3.6 (SD = 1.05). This finding corresponds with the studies conducted by Abadi et al. (2022) and Hodges (2017), which highlighted the significant impact of traditional cultural norms in Saudi Arabia on leadership views and possibilities for women. The mean score of 3.3 (SD = 0.9) regarding culture's impact on perceptions and attitudes toward leadership demonstrates significant consensus among respondents, underscoring the pivotal role of cultural frameworks in shaping leadership behavior and opportunities (van Emmerik et al., 2008).

The survey reveals mixed opinions on the influence of standardized leadership styles on gender equality in leadership positions, with a mean of 3.0 (SD = 1.1). This result indicates that while leadership styles are acknowledged as significant, there is little agreement on their function in promoting gender-inclusive leadership (Al-Shamrani, 2013). The divergent perspectives may indicate the conflict between traditional hierarchical leadership and the increasing acknowledgment of inclusive, transformational leadership styles linked to women leaders (Al-Shamrani, 2013; Górska, 2016).

A correlation study indicates significant relationships between cultural factors and gender disparities in leadership. There are substantial correlations between cultural characteristics and leadership styles (r = 0.56, p = 0.001), task assignments (r = 0.54, p = 0.003), and the proportion of women in leadership positions (r = 0.59, p < 0.001). These results highlight the significance of acknowledging cultural aspects in the analysis of leadership diversity in Saudi Arabia (Hodges, 2017; Abalkhail, 2017). Furthermore, disparities in gender representation within leadership roles were strongly correlated with leadership efficacy (r = 0.62, p < 0.001) and empowerment (r = 0.65, p < 0.001), underscoring the substantial influence of gender diversity on the perceived effectiveness and empowerment of leaders (Khan & Varshney, 2013; Jamjoom & Mills, 2022).

Regression analysis corroborates these results, demonstrating that cultural impacts on leadership attitudes (β = 0.327, p = 0.013) and leadership style $(\beta = 0.297, p = 0.009)$ are significant predictors of gender diversity in leadership roles. The model summary indicates an R-value of 0.705, indicating a moderate to high positive correlation between the independent variables and gender disparities in leadership. The R² value of 0.495 indicates that about 50% of the variance in gender gaps may be attributed to cultural and leadership characteristics, underscoring their significance in influencing gender diversity in leadership positions (Groves & LaRocca, 2011). The ANOVA analysis validated the model's robustness, shown by a statistically significant F-statistic of 8.698 (p < 0.001), hence confirming the results' validity.

Notably, interpersonal leadership abilities $(\beta=0.102, p=0.450)$ and leadership conduct $(\beta=0.112, p=0.268)$ did not have statistically significant benefits in this setting. This indicates that while these characteristics are important, they may be more shaped by overarching cultural and rather structural dynamics than independent impacts on gender gaps (McCleskey, 2014). This discovery aligns with the research of Alsubaie and Jones (2017), who emphasized that organizational frameworks and cultural conventions often surpass individual leadership abilities in influencing career progression possibilities for women.

This study's findings underscore the pressing need for organizational changes to eliminate cultural biases and leadership frameworks that hinder women's progression into leadership roles. Due to the significant association between gender diversity and leadership efficacy (Khan & Varshney, 2013), programs that foster gender-inclusive leadership need to be emphasized to improve organizational performance and leadership dynamics. Future studies should investigate the relationship between cultural views and leadership development activities, examining how these elements might be managed to promote gender equality in leadership positions within Saudi governmental organizations.

6. CONCLUSION

This research offers significant insights into the impact of cultural and organizational variables on gender diversity in leadership within Saudi governmental institutions. The results indicate that, despite continuous reforms and policy initiatives associated with Saudi Vision 2030, women are still markedly underrepresented in leadership positions, primarily due to deep-rooted cultural norms, organizational biases, and prevailing leadership

stereotypes (Syed et al., 2018; Alhassan & Al Doghan, 2022). The descriptive statistical study reveals a significant gender imbalance, with male leaders prevailing in leadership roles, while women face institutional barriers that hinder their progression. The correlation and regression analyses highlight the significant impact of cultural perceptions and leadership styles on gender diversity in leadership, emphasizing that traditional, male-dominated leadership models impede women's advancement (Hodges, 2017; Alqahtani, 2021).

The study's findings confirm the research gap, indicating that although women in leadership often display transformative and inclusive leadership styles, these attributes are often underappreciated in Saudi public organizations (Al-Shamrani, 2013). The robust association between gender diversity and leadership efficacy underscores the need for structural modifications in leadership frameworks to foster inclusive governance and equitable opportunities (McCleskey, 2014; Jamjoom & Mills, equitable 2022). Regression analysis indicates that cultural factors affecting beliefs, leadership behaviors, and organizational frameworks are substantial predictors of gender disparities in leadership, underscoring the necessity for focused interventions at both policy and institutional levels (Groves & LaRocca, 2011; Alsubaie & Jones, 2017).

Organizations must adopt proactive gender-inclusive policies to tackle these challenges, including leadership training programs for women, mentorship initiatives, and structured career progression frameworks that eradicate biases in leadership selection and promotion (Alomair, 2015; Abu Alsuood & Youde, 2018). Cultural reorientation initiatives must be implemented to confront entrenched gender prejudices and foster diversity in leadership across all organizational tiers (Al-Dajani & Alsahli, 2021; Pick, 2024).

This research enhances the discourse on gender diversity and governance in Saudi Arabia; nonetheless, its conclusions are constrained by certain constraints. The sample size, although typical of Saudi public enterprises, may not comprehensively reflect industry-specific gender discrepancies, and self-reported data may introduce response biases (Tremmel & Wahl, 2023; Tabassum & Nayak, 2021). Future research needs to implement longitudinal studies to investigate the enduring effects of gender diversity policies and analyze sector-specific leadership dynamics, especially in senior government and business roles (Arayssi, 2020; Aladwey & Alsudays, 2023).

This research underscores the essential need for gender-inclusive leadership practices in Saudi public organizations. By confronting cultural obstacles, leadership stereotypes, and systemic biases, policymakers and corporate executives can establish more equitable leadership opportunities for women, thereby aligning with Saudi Arabia's national transformation objectives for female sustainable empowerment and economic development (Jrasat & Zubaidi, 2024; Chung & Grichting Solder, 2024). The results urge companies and politicians to implement systemic change, promoting an inclusive and diverse leadership environment that maximizes the potential of both men and women in leadership positions.

Despite its strengths, this study has some limitations. The sample size may inadequately represent the intricacies of gender diversity throughout all Saudi public entities (Tabassum &

Nayak, 2021). Moreover, self-reported data may be susceptible to social desirability bias, when respondents provide responses, they consider socially acceptable instead of entirely factual (Tremmel & Wahl, 2023). The subsequent study needs to broaden sample diversity and investigate gender sector-specific obstacles to improve the generalizability of results (Al-Dajani Alsahli, 2021).

REFERENCES

- Abadi, M., Dirani, K. M., & Rezaei, F. D. (2022). Women in leadership: A systematic literature review of Middle Eastern women managers' careers from NHRD and institutional theory perspectives. Human Resource Development International, 25(1), 19-39. https://doi.org/10.1080/13678868.2020.1840847
- Abaker, M.-O. S. M., Al-Titi, O. A. K., & Al-Nasr, N. S. (2019). Organizational policies and diversity management in Saudi Arabia. *Employee Relations*, 41(3), 454–474. https://doi.org/10.1108/ER-05-2017-0104
- Abalkhail, J. M. (2017). Women and leadership: Challenges and opportunities in Saudi higher education. *Career Development International*, 22(2), 165–183. https://doi.org/10.1108/CDI-03-2016-0029
- Abdallah, A. K. (2024). The impact of gender, psychology, and cultural dimensions on leadership development in distance education. *Scientific Reports, 14*, Article 18309. https://doi.org/10.1038/s41598-024-68495-4 Abdelwahed, N. A. A., Bastian, B. L., & Wood, B. P. (2022). Women, entrepreneurship, and sustainability: The case of
- Saudi Arabia. Sustainability, 14(18), Article 11314. https://doi.org/10.3390/su141811314
- Abu Alsuood, Y., & Youde, A. (2018). An exploration of cultural factors and their influence on Saudi Arabian university deans' leadership perceptions and practices. *Education Sciences*, 8(2), Article 57. university deans' leadership percepthttps://doi.org/10.3390/educsci8020057
- Addin, E. H. M. S. (2025). Factors constraining women's participation in administrative contracts disputes arbitration: Gender equality as a sustainable development goal: A case study of Saudi Arabia. *Journal of Lifestyle and SDGs Review*, 5(1), Article e03591. https://doi.org/10.47172/2965-Review, 730X.SDGsReview.v5.n01.pe03591
- Akbar, H., Al-Dajani, H., Ayub, N., & Adeinat, I. (2023). Women's leadership gamut in Saudi Arabia's higher education sector. Gender, Work & Organization, 30(5), 1649-1675. https://doi.org/10.1111/gwao.13003
- Aladwey, L. M. A., & Alsudays, R. A. (2023). Does the cultural dimension influence the relationship between firm value and board gender diversity in Saudi Arabia, mediated by ESG scoring? Journal of Risk and Financial Management, 16(12), Article 512. https://doi.org/10.3390/jrfm16120512
- Al-Ahmadi, H. (2011). Challenges facing women leaders in Saudi Arabia. Human Resource Development
- International, 14(2), 149–166. https://doi.org/10.1080/13678868.2011.558311

 Al-Asfour, A., Tlaiss, H. A., Khan, S. A., & Rajasekar, J. (2017). Saudi women's work challenges and barriers to career advancement. Career Development International, 22(2), 184–199. https://doi.org/10.1108/CDI-11-2016-
- Al-Dajani, H., & Alsahli, M. (2021). Can women entrepreneurs thrive in Saudi Arabia? Entreprendre & Innover, 2, 49-50. https://doi.org/10.3917/entin.049.0100
- Aldekhyyel, R. N., Alhumaid, N., & Alismail, D. S. (2024). Saudi women's views on healthcare leadership in the era of Saudi 2030 health transformation. Journal of Multidisciplinary Healthcare, 17, 237-249. https://doi.org/10.2147/JMDH.S439146
 Aldossari, M., & Calvard, T. (2022). The politics and ethics of resistance, feminism and gender equality in Saudi
- Arabian organizations. Journal of Business Ethics, 181(4), 873-890. https://doi.org/10.1007/s10551-021-
- Alfarran, A., Pyke, J., & Stanton, P. (2018). Institutional barriers to women's employment in Saudi Arabia. *Equality*, Diversity and Inclusion, 37(7), 713-727. https://doi.org/10.1108/EDI-08-2017-0159
- Alhassan, S., & Al Doghan, M. A. (2022). Gender differences in leadership: Role of gender diversity, leadership style, cultural influence and leadership behavior. Croatian International Relations Review, 28(90), 332-347. https://cirrj.org/menuscript/index.php/cirrj/article/view/630
- Alhejji, H., Ng, E. S., Garavan, T., & Carbery, R. (2018). The impact of formal and informal distance on gender equality approaches: The case of a British MNC in Saudi Arabia. *Thunderbird International Business Review*, 60(2), 147-159. https://doi.org/10.1002/tie.21828
- Al-Lamky, A. (2007). Feminizing leadership in Arab societies: The perspectives of Omani female leaders. Women in Management Review, 22(1), 49-67. https://doi.org/10.1108/09649420710726229
- Almalki, O. S., Fathelrahman, A. I., Alfayez, O. M., & Bawazeer, G. A. (2024). Current situation and barriers to women's leadership in health care education in Saudi Arabia: A cross-sectional study. *Saudi* Pharmaceutical Journal, 32(5), Article 102000. https://doi.org/10.1016/j.jsps.2024.102000
- Alomair, M. O. (2015). Female leadership capacity and effectiveness: A critical analysis of the literature on higher Higher Arabia. International Journal Education, education in Saudi of https://doi.org/10.5430/ijhe.v4n4p81
- Alqahtani, T. H. (2021). Women's leadership in higher education in the Kingdom of Saudi Arabia. *Journal of Entrepreneurship & Organization Management*, 10(1), 1–6. https://www.hilarispublisher.com/open-access/womenrsquos-leadership-in-higher-education-in-the-kingdom-of-saudi-arabia-53816.html
- Alshammari, F., Pasay-An, E., Gonzales, F., & Torres, S. (2020). Emotional intelligence and authentic leadership among Saudi nursing leaders in the Kingdom of Saudi Arabia. *Journal of Professional Nursing*, 36(6), 503–509. https://doi.org/10.1016/j.profnurs.2020.04.003
- Al-Shamrani, M. S. A.-R. (2013). An empirical study of male and female leadership styles in a segregated work environment in the Kingdom of Saudi Arabia [Unpublished Ph.D. thesis, University of Hull]. https://hullrepository.worktribe.com/output/4218828/an-empirical-study-of-male-and-female-leadership-styles-in-asegregated-work-environment-in-the-kingdom-of-saudi-arabia
- Alshareef, M. N. Z., & Sandhu, K. (2015). Integrating corporate social responsibility (CSR) into corporate governance structure: The effect of board diversity and roles A case study of Saudi Arabia. *International Journal of Business and Management*, 10(7). https://doi.org/10.5539/ijbm.v10n7p1
- Alsharif, S. A. (2018). The challenges associated with women career development at the state universities in Saudi Arabia: A ground theory approach. International Journal of Gender and Women's Studies, 6(2), 18-30. https://doi.org/10.15640/ijgws.v6n2a3

- Alshdiefat, A. S., Lee, A., Sharif, A. A., Rana, M. Q., & Abu Ghunmi, N. A. (2024). Women in leadership of higher education: Critical barriers in Jordanian universities. *Cogent Education*, 11(1), Article 2357900. https://doi.org/10.1080/2331186X.2024.2357900
- Alsubaie, A., & Jones, K. (2017). An overview of the current state of women's leadership in higher education in Saudi Arabia and a proposal for future research directions. *Administrative Sciences*, 7(4), Article 36. https://doi.org/10.3390/admsci7040036
- Alwazzan, L., & Rees, C. E. (2016). Women in medical education: Views and experiences from the Kingdom of Saudi Arabia. Medical Education, 50(8), 852–865. https://doi.org/10.1111/medu.12988
- Alzaaqi, S. M., Sheerah, H. A., Arafa, A., Alqahtani, D. M., Alqadi, S. A., Alsalamah, H. A., Ismail, E. H., Nouh, M. A., & AlSalamah, S. A. (2025). Empowering women in the Saudi health sector: Challenges, opportunities, and policy interventions. *Iournal of Epidemiology and Global Health*, 15(1), Article 22. policy interventions. Journal of Epidemiology https://doi.org/10.1007/s44197-025-00361-6
- Angelakis, A., Inwinkl, P., Berndt, A., Ozturkcan, S., Zelenajova, A., & Rozkopal, V. (2024). Gender differences in leaders' crisis communication: A sentiment-based analysis of German higher education leaderships' online posts. Studies in Higher Education, 49(4), 609-622. https://doi.org/10.1080/03075079.2023.2246505
- Arayssi, D. (2020). The socio-economic development of the Kingdom of Saudi Arabia: Gender Inequality Index and policy recommendation. https://doi.org/10.2139/ssrn.3726774

 Badura, K. L., Grijalva, E., Newman, D. A., Yan, T. T., & Jeon, G. (2018). Gender and leadership emergence: A meta-analysis and explanatory model. Personnel Psychology, 71(3), 335–367. https://doi.org/10.1111 /peps.12266 Bakr, M. B. B. (2022). Women and leadership: Challenges to women empowerment in academic leadership roles in
- Arabia. Positive Psychology Wellbeing, Journal of and 6(2),https://journalppw.com/index.php/jppw/article/view/11656
- Billah, W., & Kabir, S. L. (2024). Role of women in civil service: Implications for effective and inclusive governance in Bangladesh. In H. Zafarullah & F. Azmat (Eds.), *Governance and sustainable development in South Asia: Bridging the gap* (pp. 107–122). Springer Cham. https://doi.org/10.1007/978-3-031-67616-1_6
- Burke, S., & Collins, K. M. (2001). Gender differences in leadership styles and management skills. *Women in Management Review*, 16(5), 244–256. https://www.deepdyve.com/lp/emerald-publishing/genderdifferences-in-leadership-styles-and-management-skills-l6YWavygXR
- Chalermchaikit, V., Kozak, M., & Preudhikulpradab, S. (2024). Gender inclusion: The practices of organizational development and human resource management. *International Journal of Hospitality Management*, 120, Article 103773. https://doi.org/10.1016/j.ijhm.2024.103773
- Chebbi, K., & Ammer, M. A. (2022). Board composition and ESG disclosure in Saudi Arabia: The moderating role of
- corporate governance reforms. *Sustainability*, *14*(19), Article 12173. https://doi.org/10.3390/su141912173 Chin, J. L. (2013). Diversity leadership: Influence of ethnicity, gender, and minority status. *Open Journal of Leadership*, *2*(1), 1-10. https://doi.org/10.4236/ojl.2013.21001
- Chung, R. Y.-A., & Grichting Solder, A. (2024). Gender-inclusive urban planning: European frameworks as a potential methodology for Arabian Gulf urban parks. *Archnet-IJAR*, 18(3), 519–540. https://doi.org/10.1108/ARCH-09-2023-0260
- Dahlan, D. A. (2024). Who is the next leader? Understanding women leadership development and succession planning in Saudi Arabian higher educational institutions. Frontiers in Sociology, 9, Article 1442543. https://doi.org/10.3389/fsoc.2024.1442543
- Doumato, E. A. (1992). Gender, monarchy, and national identity in Saudi Arabia. British Journal of Middle Eastern Studies, 19(1), 31-47. https://doi.org/10.1080/13530199208705547
 Ehrhart, M. G., & Klein, K. J. (2001). Predicting followers' preferences for charismatic leadership: The influence of
- follower values and personality. The Leadership Quarterly, 12(2), 153-179. https://doi.org/10.1016/S1048-9843(01)00074-1
- Eichenauer, C., & Ryan, A. M. (2024). Gender and leadership evaluations: What happens if we actually measure expectations-behavior incongruence? Gender in Management, 39(5), 664-679. https://doi.org/10.1108/GM-06-2023-0205
- Eklund, K. E., Barry, E. S., & Grunberg, N. E. (2017). Gender and leadership. In A. Alvinius (Ed.), Gender differences in different contexts (pp. 129-150). IntechOpen. https://doi.org/10.5772/65457
- Ely, R. A., & Rhode, D. L. (2010). Women and leadership: Defining the challenges. In N. Nohria & R. Khurana (Eds.), Handbook of leadership theory and practice. Harvard Business Publishing. https://www.hbsp.harvard.edu /product/6144BC-PDF-ENG
- Gooty, J., Ruggs, E. N., Aguinis, H., Bergeron, D. M., Eby, L. T., van Knippenberg, D., Post, C., Rupp, D. E., Thatcher, S. M. B., Tonidandel, S., & Yammarino, F. J. (2023). Stronger together: A call for gender-inclusive leadership in business schools. *Journal of Management*, 49(8), 2531–2540. https://doi.org/10.1177 /01492063231178792
- Górska, Á. (2016). Gender differences in leadership. Studia i Materiały Wydziału Zarządzania Uniwersytet Warszawski, 2016(1), 136-144. https://doi.org/10.7172/1733-9758.2016.20.10
- Groves, K. A., & LaRocca, M. A. (2011). An empirical study of leader ethical values, transformational and transactional leadership, and follower attitudes toward corporate social responsibility. Journal of Business
- Ethics, 103, 511-528. https://doi.org/10.1007/s10551-011-0877-y Guy, M. E., Newman, M. A., & Mastracci, S. H. (2008). Emotional labor: Putting the service in public service. Routledge. Hodges, J. (2017). Cracking the walls of leadership: Women in Saudi Arabia. Gender in Management, 32(1), 34-46. https://doi.org/10.1108/GM-11-2015-0106
- Hogan, S. J., & Coote, L. V. (2014). Organizational culture, innovation, and performance: A test of Schein's model. Journal of Business Research, 67(8), 1609–1621. https://doi.org/10.1016/j.jbusres.2013.09.007
- Hoyt, C. L., & Simon, S. (2024). Social psychological approaches to women and leadership theory. In S. R. Madsen (Ed.), *Handbook of research on gender and leadership* (pp. 65–83). Edward Elgar Publishing. https://doi.org/10.4337/9781035306893.00015
- Husu, L., Hearn, J., Lämsä, A.-M., & Vanhala, S. (Eds.). (2010). *Leadership through the gender lens: Women and men in organizations* (Hanken School of Economics Research Report). https://helda.helsinki.fi/items/8ffd14dec339-447e-bddb-8f042957967a
- & Fang, H.-X. (2019). The impact of board gender diversity on corporate social responsibility in the Arab Gulf states. Gender in Management, 34(7), 577-605. https://doi.org/10.1108/GM-07-2018-0087
- Itzkovitch-Malka, R. (2024). How do gender stereotypes about leadership positions change in the face of a crisis? Research & Politics, 11(4). https://doi.org/10.1177/20531680241293602

- Jamjoom, L. A., & Mills, A. J. (2022). Narratives of workplace resistance: Reframing Saudi women in leadership. Human Relations, 76(7), 955-989. https://doi.org/10.1177/00187267221087593
- Jizi, M., Nehme, R., & Melhem, C. (2022). Board gender diversity and firms' social engagement in the Gulf Cooperation Council (GCC) countries. *Equality, Diversity and Inclusion*, 41(2), 186–206. https://doi.org/10.1108/EDI-02-2021-0041
- Jrasat, S. S., & Zubaidi, F. R. F. (2024). Saudi women empowerment in leadership positions in health care sector: Barriers to their success. *International Journal of Professional Business Review*, *9*(4), Article e03828. https://doi.org/10.26668/businessreview/2024.v9i4.3828
- Kamalizeni, C., Hoque, M., Kader, A., & Mapudzi, H. (2018). Gender inclusive leadership: Exploring women underrepresentation in public enterprises — The case of Swaziland. Journal of Public Administration, 53(4), 849-872. https://journals.co.za/doi/epdf/10.10520/EJC-1723ffe270
- Khan, S. A., & Varshney, D. (2013). Transformational leadership in the Saudi Arabian cultural context: Prospects and challenges. In J. Rajasekar & L.-S. Beh (Eds.), Culture and gender in leadership: Perspectives from the Middle East and Asia (pp. 200-227). Palgrave Macmillan London. https://doi.org/10.1057/9781137311573_11
- Lewis, A. E., & Fagenson-Eland, E. A. (1998). The influence of gender and organization level on perceptions of leadership behaviors: A self and supervisor comparison. *Sex Roles*, *39*(5), 479–502. https://doi.org/10.1023/A:1018831328037
- Maher, K. J. (1997). Gender-related stereotypes of transformational and transactional leadership. Sex Roles, 37(3), 209-225. https://doi.org/10.1023/A:1025647811219
- McCleskey, J. A. (2014). Situational, transformational, and transactional leadership and leadership development. *Journal of Business Studies Quarterly*, 5(4), 117–130.
- Najmaei, A. (2018). Revisiting the strategic leadership paradigm: A gender inclusive perspective. In S. Adapa & A. Sheridan (Eds.), *Inclusive leadership: Negotiating gendered spaces* (pp. 203–228). Palgrave Macmillan Cham. https://doi.org/10.1007/978-3-319-60666-8_9
- Otayf, A. A. M. (2024). Women in leadership: Exploring the barriers and opportunities of academic Saudi women's subjective experiences in leadership roles in Saudi government universities [Doctoral dissertation, University of St. Thomas]. https://researchonline.stthomas.edu/esploro/outputs/doctoral/Women-in-Leadership-Exploring-the-Barriers/991015284334703691
- Ottmann, D. A. (2024). Our common Gulf cities: Agenda for equitable AEC industries for sustainable urban development. *Archnet-IJAR*, 18(3), 672–690. https://doi.org/10.1108/ARCH-11-2023-0295
 Parveen, M. (2023). Reflection of Saudi women's participation and leadership: A study on the gender differences in
- leadership and structural barriers. Rupkatha Journal on Interdisciplinary Studies in Humanities, 15(2), 1-12. https://doi.org/10.21659/rupkatha.v15n2.16
- Pick, K. (2024). Women, leadership, and power. In S. R. Madsen (Ed.), Handbook of research on gender and leadership (pp. 220-240). Edward Elgar Publishing. https://doi.org/10.4337/9781035306893.00025
- Proff, A., & Musalam, R. (2024). Institutional agents: A significant facilitator of the advancement of women leaders. Gulf Education and Social Policy Review, 5(2), 64–92. https://doi.org/10.18502/gespr.v5i2.15184
 Purwaningtias, G., & Purba, D. E. (2025). The association between inclusive leadership and pro-social rule breaking:
- Gender as moderator. Journal of Gender Equality Disability Social Inclusion and Children, 2(2), 71-81. https://doi.org/10.61511/jgedsic.v2i2.2025.1458
- Qureshi, M. İ., Zaman, K., & Bhatti, M. N. (2011). The impact of culture and gender on leadership behavior: Higher education and management perspective. https://doi.org/10.5267/j.msl.2011.05.008 Management Science Letters, 531-540.
- https://doi.org/10.526//J.msi.2011.05.006

 Riddle, L., & Ayyagari, M. (2011). Contemporary Cleopatras: The business ethics of female Egyptian managers.

 Therefore Projects and Society Contemporary Middle Fastern Issues, 4(3), 167-192. *Education, Business and Society: Cor* https://doi.org/10.1108/17537981111159957
- Rincón, V., González, M., & Barrero, K. (2017). Women and leadership: Gender barriers to senior management positions. *Intangible Capital*, 13(2), 319–386. https://doi.org/10.3926/ic.889
- Sabharwal, M., Levine, H., & D'Agostino, M. J. (2017). Gender differences in the leadership styles of MPA directors.
- Journal of Public Affairs Education, 23(3), 869–884. https://doi.org/10.1080/15236803.2017.12002293
 Setyaningrum, G., & Juansih. (2024). Gender-based challenges in women's leadership careers: A literature synthesis.

 Technium Social Sciences Journal, 63(1), 252–267. https://doi.org/10.47577/tssj.v63i1.11929
 Sian, S., Agrizzi, D., Wright, T., & Alsalloom, A. (2020). Negotiating constraints in international audit firms in Saudi
- Arabia: Exploring the interaction of gender, politics, and religion. *Accounting, Organizations and Society, 84,* Article 101103. https://doi.org/10.1016/j.aos.2020.101103
- Sobaih, A. E. E., & Abu Elnasr, A. E. (2024). Challenges to cracking the glass ceiling among Saudi women in the tourism industry. *Tourism and Hospitality*, *5*(1), 203–224. https://doi.org/10.3390/tourhosp5010014
 Song, Z. (2024). Literature review of women's leadership: A cross-cultural comparative study of gender equality.

 **Advances in Applied Sociology, 14(8), 413–418. https://doi.org/10.4236/aasoci.2024.148029
- Syed, J., Ali, F., & Hennekam, S. (2018). Gender equality in employment in Saudi Arabia: A relational perspective. Career Development International, 23(2), 163-177. https://doi.org/10.1108/CDI-07-2017-0126
- Tabassum, N., & Nayak, B. S. (2021). Gender stereotypes and their impact on women's career progressions from a managerial perspective. *IIM Kozhikode Society & Management Review*, 10(2), 192-208. a managerial perspective. *IIM Kozhikode Society & Management Review*, 10(2), 192-208. https://doi.org/10.1177/2277975220975513

 Thelma, C. C., & Ngulube, L. (2024). Women in leadership: Examining barriers to women's advancement in leadership.
- positions. *Asian Journal of Advanced Research and Reports*, *18*(6), 273-290. https://doi.org/10.9734/ajarr/2024/v18i6671

 Tremmel, M., & Wahl, I. (2023). Gender stereotypes in leadership: Analyzing the content and evaluation of stereotypes about typical, male, and female leaders. *Frontiers in Psychology*, *14*, Article 1034258.
- https://doi.org/10.3389/fpsyg.2023.1034258
 ur Rehman, A., & Alorifi, A. M. (2024). Shattering the glass ceiling: A study on the preferred leadership style of Saudi
- women entrepreneurs. *Calitatea*, *25*(199), 279–290. https://doi.org/10.47750/QAS/25.199.31 van Emmerik, IJ. H., Euwema, M. C., & Wendt, H. (2008). Leadership behaviors around the world: The relative importance of gender versus cultural background. *International Journal of Cross Cultural Management,* 8(3), 297–315. https://doi.org/10.1177/1470595808096671
 Xu, C., Mussagulova, A., & Zhou, Q. (2024). The role of gender equality in advancing development. *Public*
- Administration and Development, 44(4), 252-261. https://doi.org/10.1002/pad.2042